Author's response to reviews

Title: Comparative analysis of the human hepatic and adipose tissue transcriptomes during LPS-induced inflammation leads to the identification of differential biological pathways and candidate biomarkers

Authors:

Ewa Szalowska (ewa.szalowska@wur.nl)
Martijn Dijkstra (m.dijkstra@rug.nl)
Marieke G.L. Elferink (M.G.L.Elferink@rug.nl)
Desiree Weening (d.weening@med.umcg.nl)
Marcel de Vries (marcel.de.vries@med.umcg.nl)
Marcel Bruinenberg (m.bruinenberg@lifelines.umcg.nl)
Annemieke Hoek (a.hoek@og.umcg.nl)
Han Roelofsen (j.roelofsen@med.umcg.nl)
Geny M.M. Groothuis (g.m.m.groothuis@rug.nl)
Roel J. Vonk (r.j.vonk@med.umcg.nl)

Version: 8 Date: 22 August 2011

Author's response to reviews:

Dear BMC Series Editorial Production Team,

Please find a revised version of the manuscript. The changes were made accordingly to suggestions in the previous email.

Regards,

Ewa